Author's response to reviews

Title: Melatonin prevents morphine-induced hyperalgesia and tolerance in rats: role of protein kinase C and N-methyl-D-aspartate receptors

Authors:

Li Song (huaxisongli@hotmail.com)
Chaoran Wu (chaoranwu@aliyun.com)
Yunxia Zuo (zuoyunxiahxa@qq.com)

Version: 2
Date: 30 October 2014

Author's response to reviews: see over
Dear editor:

Thank you very much for the email and your comments. We greatly appreciated that you give us this opportunity to revise the manuscript.

We have revised this manuscript according to your comments. And Changes have been made in the revised manuscript.

Point-by-point response to each comments

**Line numbering:**

Please revise your manuscript to include line and page numbers. Authors are asked to ensure that line numbering is included in the main text file of their manuscript at the time of submission to facilitate peer-review.

**Response:** We have revised our manuscript to include line and page numbers.

**List of abbreviations**

If abbreviations are used in the text they should be defined in the text at first use, and a list of abbreviations can be provided, which should precede the competing interests and authors' contributions.

**Response:** We have added the abbreviations in the manuscript and the abbreviations were precede the competing interests and authors' contributions.

Many thanks.

Best wishes,

Yunxia Zuo, M.D., Ph.D.
Translational Medical Neuroscience Center, and Department of Anesthesiology
West China Hospital
Sichuan University
Chengdu 610041, China